\documentstyle[a4]{article}

\begin{document}

\title{A bibliography on coordination}
\author{Paolo Ciancarini\\
Dipartimento di Scienze dell'Informazione\\
University of Bologna - Italy\\
Email: {\tt ciancarini@cs.unibo.it}}


\maketitle

This is a list of papers related to coordination models, 
languages, architectures, and applications.


\section{General}
In this section we list papers, books, and conference proceedings 
concerning the general definition of coordination models and 
languages and related issues
\cite{AHL96,Cia90d,CarGel90b,CarGel92a,Adl95a,Cia96c,MalCro94,CiaHan96,GarLeM97,Weg97,Cia97a,CHT98,CiaTol98,PapArb98a,CiaWol99,PorRom00,CiaWoo00}.

\bibliographystyle{abbrv}    
\bibliography{biblio}

\end{document}
